Author’s response to reviews

Title: Community initiated kangaroo mother care and early child development in low birth weight infants in India-A randomized controlled trial

Authors:

Sunita Taneja (sunita.taneja@sas.org.in)
Bireshwar Sinha (bireshwar.sinha@sas.org.in)
Ravi Prakash Upadhyay (ravi.upadhyay@sas.org.in)
Sarmila Mazumder (sarmila.mazumder@sas.org.in)
Halvor Sommerfelt (Halvor.Sommerfelt@uib.no)
Jose Martines (martinesjosec@gmail.com)
Suresh Kumar Dalpath (sureshdalpath@yahoo.com)
Rakesh Gupta (rguptiaias@gmail.com)
Patricia Kariger (patriciakariger@gmail.com)
Rajiv Bahl (bahr@who.int)
Nita Bhandari (nita.bhandari@sas.org.in)
Tarun Dua (duat@who.int)

Version: 2 Date: 16 Mar 2020

Author’s response to reviews:

We have modified the manuscript according to the changes suggested. We would like to thank the reviewers and the Editor for their valuable suggestions.